Author's response to reviews

Title: Molecular characterization of influenza viruses collected from young children in Uberlandia, Brazil - from 2001 to 2010: presence of reassortment between 2002 strains

Authors:

Thelma F M S Oliveira (thelmao@umuarama.ufu.br)
Jonny Yokosawa (iyokosawa@icbim.ufu.br)
Fernando C Motta (fcm@ioc.fiocruz.br)
Marilda M Siqueira (mmsiq@ioc.fiocruz.br)
Hélio L Silveira (heliols@yahoo.com.br)
Divina A O Queiróz (queiroz.divina@gmail.com)

Version: 5 Date: 30 October 2014

Author's response to reviews:

Dear Dr. Nathaniel Nazareno,

The abstract and the main manuscript text were revised and modified according to the guidelines.

Best regards,

Thelma Oliveira